Author’s response to reviews

Title: Making progress: the role of Cancer Councils in Australia in Indigenous cancer control

Authors:

Sandra C Thompson (sandra.thompson@cucrh.uwa.edu.au)
Shaouli Shahid (shaouli.shahid@uwa.edu.au)
Michelle DiGiacomo (michelle.digiacomo@uts.edu.au)
Leanne Pilkington (leanne.pilkington@health.wa.gov.au)
Patricia M Davidson (patriciamary.davidson@uts.edu.au)

Version: 4 Date: 4 April 2014

Author’s response to reviews:

I have completed the 2 minor discretionary revisions raised by the reviewer

- Additional references to the previous scan methods have been added.
- The sentence identified has been reworded to be clearer.